

Received March 1, 2020, accepted March 22, 2020, date of publication April 1, 2020, date of current version April 16, 2020. *Digital Object Identifier 10.1109/ACCESS.2020.2984925*

A Survey of Voice Pathology Surveillance Systems Based on Internet of Things and Machine Learning Algorithms

FAHAD TAHA AL-DHIEF¹, (Student Member, IEEE),

NURUL MU'AZZAH ABDUL LATIFF^{®1}, (Senior Member, IEEE), NIK NOORDINI NIK ABD. MALIK¹, (Member, IEEE), NASEER SABRI SALIM², MARINA MAT BAKI³, MUSATAFA ABBAS ABBOOD ALBADR⁴, AND MAZIN ABED MOHAMMED^{®5}

¹Faculty of Engineering, School of Electrical Engineering, Universiti Teknologi Malaysia (UTM), Johor Bahru 81310, Malaysia

²Computing and Information Technology, Sohar University, Sohar 311, Oman

³Department of Otorhinolaryngology, Faculty of Medicine, Universiti Kebangsaan Malaysia Medical Centre, Kuala Lumpur 56000, Malaysia

⁴CAIT, Faculty of Information Science and Technology, Universiti Kebangsaan Malaysia, Bangi 43600, Malaysia

⁵College of Computer Science and Information Technology, University of Anbar, Ramadi 45654, Iraq

Corresponding author: Nurul Mu'azzah Abdul Latiff (nurulmuazzah@utm.my)

This work was supported in part by the Ministry of Higher Education, Malaysia, and in part by the Universiti Teknologi Malaysia (UTM) under Research Grant R.J130000.2651.18J74.

ABSTRACT The incorporation of the cloud technology with the Internet of Things (IoT) is significant in order to obtain better performance for a seamless, continuous, and ubiquitous framework. IoT has many applications in the healthcare sector, one of these applications is voice pathology monitoring. Unfortunately, voice pathology has not gained much attention, where there is an urgent need in this area due to the shortage of research and diagnosis of lethal diseases. Most of the researchers are focusing on the voice pathology and their finding is only to differentiating either the voice is normal (healthy) or pathological voice, where there is a lack of the current studies for detecting a certain disease such as laryngeal cancer. In this paper, we present an extensive review of the state-of-the-art techniques and studies of IoT frameworks and machine learning algorithms used in the healthcare in general and in the voice pathology surveillance systems in particular. Furthermore, this paper also presents applications, challenges and key issues of both IoT and machine learning algorithms in the healthcare. Finally, this paper highlights some open issues of IoT in healthcare that warrant further research and investigation in order to present an easy, comfortable and effective diagnosis and treatment of disease for both patients and doctors.

INDEX TERMS Internet of Things, machine learning algorithms, the healthcare sector, voice pathology surveillance systems.

I. INTRODUCTION

The healthcare sector is considered as one of the hottest applications areas in the IoT, where IoT has the possibility to enhance several of the medical applications such as elderly care, remote health monitoring, chronic diseases, and fitness programs. Therefore, many medical devices, diagnostic and imaging equipment and sensors can be seen as intelligent devices that are part of the essential elements in IoT [1]. Also, it is expected that the cost of the IoT-based healthcare services can be reduced and consequently the quality of life can be enhanced and enriches the experience

The associate editor coordinating the review of this manuscript and approving it for publication was Muhamamd Aleem^(D).

of the users. From the viewpoint of the healthcare suppliers, the IoT has the possibility to minimize the downtime of the devices through the remote provision [2]. Moreover, the IoT presents the effective scheduling of limited resources by ensuring that their best service can be utilized by more patients. Fig. 1 illustrates the recent healthcare trends in IoT technologies. Ease of cost-efficient interactions during seamless connectivity across the patients, healthcare organizations and clinics are significant trends. The modern wireless technologies in the healthcare networks are expected to support early diagnosis, medical emergencies, chronic diseases and real-time monitoring [3], [4]. At this stage, a comprehensive understanding of present research in the IoT with respect to healthcare applications is expected to



FIGURE 1. Healthcare trends in IoTs.

be beneficial for several stakeholders interested in further research.

Alongside the services of IoT in healthcare applications, machine learning offers tools, methods, and techniques that can assist in solving prognostic and diagnosis problems in different kinds of medical domains. Machine learning has been utilized in the analysis of clinical parameters and their combinations for prediction such as extraction of the knowledge of the medical for outcome research, the comprehensive management for the patient, treatment planning and support, and the prediction of illness progress [5], [6]. Integration of machine learning within the healthcare sector provides opportunities to ease and develop the medical expert's work and enhance the quality and efficiency of medical care [7].

The merging between the cloud and IoT can provide a broad application in social and daily life in general and in the healthcare domain in particular. This is due to the importance of this domain, where the healthcare applications can thrive through the adoption of the cloud and IoT model in the healthcare field which can lead to bringing various opportunities to medical IoT. For example, continuous monitoring applications where there are numerous cases of patients that require long-term monitoring such as a patient who has chronic disease [8], [9]. With regards to this, providing continuous monitoring is a vital issue. Moreover, the follow-up procedure is very important for a patient to optimize diagnosis and recovery time. For instance, to check the abnormal growth of the vocal folds and the quality of the voice.

In the healthcare sector, the diagnosis of some diseases is possible using certain features of speech signals [10]. Due to the nature of work and unhealthy social habits, certain people are prone to the risk of voice problem where the speech signal of pathological voice has become an important topic in this field [11]–[13].

Thus, in addition to IoT and machine learning, this paper also focuses on the implementation of these technologies in voice pathology field. Although voice pathology is a very important area, it has not gained much attention, and hence, there is an urgent need in this area due to the shortage of researches and diagnosis of the diseases by voice. One of these diseases is laryngeal cancer, where this cancer is one of the most common head and neck cancer worldwide [14]. This disease starts with the voice box problem or also known as larynx. Nowadays, laryngeal cancer is the most dangerous disease that affects the patient's voice and can lead to death [15]. Thus, a new and effective framework or technique to detect this cancer via patient's voice at an early stage is crucial, where there is a good chance of cure if this cancer is treated in the early stage. Regrettably, most of the researchers and developers are focusing on the voice pathology and their finding is only to differentiating either the voice is normal (healthy) or pathological voice such as in [16]–[19], where there is no sharp detection for diseases based on voice abnormality. Also, for machine learning algorithms, these technologies suffer from low prediction accuracy rate and time-consumption in the pathology monitoring approaches. Nevertheless, in this paper, we present an inclusive survey of recent studies and approaches for both IoT technologies and machine learning algorithms in the healthcare field in general, and in voice pathology area in particular. Also, this paper provides several major challenges which should be taken into account for future research.

The rest of the paper is organized as follows. Section II presents a brief concept of IoT. Section III presents the role of IoT in the healthcare area as well as IoT applications, IoT challenges and IoT voice pathology. Section IV discusses the state-of-the-art of IoT in the healthcare sector. Section V gives the concept of the machine learning. Section VI describes the key role of machine learning involved in the healthcare area, its applications, identifies many of the machine learning challenges and issues in healthcare and provides the implementation of machine learning in voice pathology area. Section VII discusses the recent studies and methods of machine learning algorithms used in the healthcare sector. Finally, Section VIII concludes the paper.

II. INTERNET OF THINGS (IOT)

The IoT is a term that was proposed by Kevin Ashton in 1999 [20]. IoT has gained more and more attention from academic researchers, government, and industry all over the world, wherein the IoT concept has become popular. IoT can be referred to all things (or objects) that are connected to the Internet via sensor devices. The Auto Identification Center (AIDC) is a center for technologies of radio frequency identification and wireless sensor network, where AIDC has played the main role to materialize the term of IoT [21]. There are several definitions of IoT that have reappeared, and the term is progressively evolving as technology advances and various ideas move forward [22]. The IoT expands into our daily lives over a wireless network of objects that can be uniquely identified [23]. One of the newest and important techniques in operating methods is Radio Frequency Determination (RFID). RFID is an innovative technology that once again utilizes the waves of radio to transmit data through an electronic tag, labelled, and linked to an object, via the reader for the objective of tracking and identifying the object.

RFID system consists of three parts: scanning antenna and transceiver (often combined into one as RFID reader) as well as central computer system [24], [25]. The RFID tag stores all information, and standard elements. Through the wireless network, the tags are conveyed to the central computer system. The object can be identified via the private flag. At the same time, these tags are able to be shared across the Internet in order to manage the object. Since the IoT has been invented in 1999 [22], it has been attributed to the variety of descriptions for such a network, where it has been described also as a paradigm network [26], [27], a concept [28], an Internet application [29] and a global network infrastructure [30]. Moreover, in the IoT phrase, the word 'Things' has been replaced with different alternative terms such as Internet of Everything (IoE) [31], Internet of Anything (IoA) [32], Internet of People (IoP) [33], [34], and the Internet of Signs (IoS) [35].

In overall, the Internet represents a common network, diffused across a broad geographical area. This network can be managed and shared through different protocols, known devices and connectors such as HTTP, HTTPS, modern computers, routers, switches, Bluetooth, Wi-Fi technology, fiber cables, Ethernet, and different personal computers, tablets, and smartphones [36].

III. IOT IN THE HEALTHCARE SECTOR

The healthcare sector is considered as one of the topmost challenges that are faced by every country nowadays. Although the healthcare industry invests heavily in information technology, organizations in healthcare nowadays still depend on the paper medical records and doctor's handwriting notes to relay information about the patients. This makes the sharing of patients' data between departments and clinicians are complex and limited, where doctors obtain information only by a physical assessment during their patient's visit to the hospital. On the other hand, using the IoT in healthcare can provide the doctor to be accessible to the patient's record easily, and at any time, besides being able to track the patient's condition in real-time. The cooperation between the IoT with the cloud in the healthcare field can lead to a better organization of the healthcare sector. In particular, the management of clinical services and patient's data will be more efficient. Moreover, public health surveillance, treatment and diagnostics can be done in a more convenient, trustable and efficient manner with less cost. IoT technology can also provide services such as online interaction with patients, tracking patients' condition and doctor's locations, and tracking medical reports for the patients [37].

IoT has the ability to connect different objects such as P2D (Patient-to-Doctor), S2M (Sensor-to-Mobile), D2M (Device-to-Machine), M2H (Mobile-to-Human), D2M (Doctor-to-Machine), O2O (Object-to-Object), T2R (Tag-to-Reader), and P2M (Patient-to-Machine). It connects smart devices, humans, dynamic systems and machines which provide and ensure an effective healthcare monitoring system. However, patients monitoring represents one of the biggest challenges



FIGURE 2. IoT devices in healthcare.

in research directions using IoT in the healthcare field. The US Institute of Medicine revealed that medical errors remain despite advances in healthcare technology, where 3 persons die out of 400,000 people per year because of these errors. The main reason for these errors is due to failures in running proper tests or no follow up with doctors, delayed in the diagnosis, and disability to access patient's medical history [38]. IoT can support potentially life-saving applications within the healthcare industry by collecting data from the devices, showing the patient information, and diagnosing in real-time the entire system of patient care [39]. Fig. 2 shows some of IoT applications in the healthcare sector, where IoT devices can be used for different purposes such as medication reminder, heartbeat sensing and monitoring, and blood pressure measurement. Nowadays, there are many healthcare devices that operate in all over the world, and since these devices are related with people health, the diagnostic accuracy and data security should be highly effective, trusted and secured. Hence, diagnosis results can be dependable and proper treatment can be given to the patient. Using IoT technology, doctors or caregivers have the capability to efficiently manage and monitor patient health and can economize precious minutes each day. Using IoT, there is no need to physically visit the patients, where caregivers or doctors can provide a remote tracking and diagnosis for the patients. Using Wi-Fi and sensors in the hospital, the right department can be determined when retrieving the sensed information [40].

In terms of clinical care, any patients who require close attention or non-invasive monitoring due to their physiological status can be continuously monitored using the IoT-driven sensor. The sensor collects the physiological information from the patient to be analyzed and uses gateways to further transmit the information. The obtained information will be stored in the cloud. This information is then sent to the caregivers/doctors wirelessly for further analysis as shown in Fig. 3. Consequently, this improves the care quality and further decreases the cost of treatment for the patient.

Meanwhile, a remote health monitoring system that is based on IoT can track a patient's vital signals in real-time and responds if there is any problem in the patient's health. A sensor device can be attached to the patient as shown



FIGURE 3. Clinical care system for constantly monitoring the physiological status.



FIGURE 4. The remote health monitoring system.

in Fig. 4. It transmits data about the vital signs from the place where the patient is located. Through the telecom network, the transmitter is connected to the hospital [41]. The hospital system monitors and reads the patient's vital signs remotely. In the same way, when the sensor is implanted into the patient's body, the data can be electronically transmitted. The information which is transmitted will be securely sent to healthcare providers/caregivers.

A. IoT APPLICATIONS IN HEALTHCARE SECTOR

IoT technique plays a significant role in many applications and different aspects of the healthcare sector. IoT frameworks/techniques have been applied in several healthcare applications such as blood pressure monitoring to present an easy and comfortable way for patients [42], rehabilitation systems for patients after a certain disease [43], oxygen saturation monitoring to monitor the patient health condition [44], wheelchair management to interact with surroundings by the elderly and disabled people [45], and smartphones that have healthcare apps to present remote medical consulting and many services [46]. Moreover, IoT techniques present lowcost healthcare services for patients and high-quality drugs management and medication adherence. Thus, these IoT based applications can also reduce the error rate and improve the treatment outcomes of the patient. Brief explanations of specific healthcare applications based on IoT are given as follow:

1) BLOOD PRESSURE MONITORING

Blood pressure (BP) is one of the most important physiological parameters of the human body. Nowadays, the use of safe and simple blood pressure monitor at home has become common [42]. An electronic blood pressure monitor that is connected to the IoT sensor can collect real-time information of the patient's blood pressure level. This information can then be relayed to the doctors/caregivers via IoT system.

2) REHABILITATION SYSTEM

This system can improve and restore the functional capabilities and also enhance the quality of life for the people who are suffering from some disabilities in terms of mitigating problems which are related with ageing populations and when there is a lack of health experts [43]. An automated design method is proposed in [47] for IoT-based intelligent rehabilitation methods. This automated design has demonstrated that the IoT platform can connect effectively to all needed resources to present real-time information interactions. IoT technologies can form a valuable infrastructure in comprehensive rehabilitation to support remote consultation [48]. There are several rehabilitation methods which are based on IoT technologies. For instance, rehabilitation training method of hemiplegic patients [49], and medical rehabilitation method for a smart city [50].

3) OXYGEN SATURATION MONITORING

The pulse oximeter is a device which continuously monitors the blood oxygen saturation of the patient in a non-invasive way [44]. There are many advances in communication technology, for instance, wireless networks, and medical sensors which are booming at present days because of the low power consumption and low loss. The continuous monitoring using pulse oximeters are applied in many medical applications in order to know the oxygen levels in the blood and also the heart rate. The IoT sensor which is connected to the patient's body will monitor and sense the patient's heart rate and oxygen levels and hence can guide the patient to limit his/her physical activity [51].

4) WHEELCHAIR MANAGEMENT

Wheelchairs are normally used by the people who are suffering from physical illness or any other physical disabilities. Wheelchairs can be benefited from the IoT technology. Wireless Body Area Networks (WBANs) that can connect smart objects with the Internet can also be used as a people-centric sensing device for wheelchair users. For example, pressure cushion sensors (a type of resistive pressure sensor) that are attached to the wheelchair are able to detect movement such as when a person sitting on it falls down from the wheelchair. A smart wheelchair can also be equipped with another accelerator sensor which detects the falling of the wheelchair [45]. Consequently, the caregiver of a person using the wheelchair can continuously take care of his/her patient remotely without the need to be near the patient at all the time.

5) HEALTHCARE SOLUTIONS USING SMARTPHONES

Mobile devices and apps of healthcare offer numerous benefits for Health Care Professionals (HCPs). There are many medical healthcare applications which are now available in many ways and ready to be accessed such as health record, information and time, communication and consulting with doctors, patient continuous monitoring and proper clinical decision making [46]. With the use of smartphone apps and sensors, the point of care and the access to care have been increased and this supports the improvement of patient's outcomes.

B. CHALLENGES OF IOT IN HEALTHCARE SECTOR

Numerous researchers have worked on executing and designing different IoT-based healthcare services, and they intend to solve different architectural and technological problems related to these services. Nevertheless, there are many other open issues and challenges in IoT-based healthcare that require thorough investigation and discussion. These issues can be summarized as follows:

1) COST ANALYSIS

It has been acknowledged by researchers that IoT is a lowcost technology but unfortunately, there is no comparative study to present any evidence of this. Thus, cost analysis in particular in the healthcare field can be beneficial.

2) THE PROCESS OF APP DEVELOPMENT

When developing an app on the Android platform, there are four primary steps: the setup, development, debugging and testing, and publishing. Similar approaches are generally taken on other platforms. In the process of healthcare app development, the participation of an authorized body or association of medical experts is typically required to ensure an app of acceptable quality. In addition, regular updates on healthcare apps to keep abreast recent advances in medical science are vital.

3) TECHNOLOGY DEVELOPMENT

The organizations of the healthcare can improve their current devices and sensors across the healthcare field for smart resources by incorporating IoT approaches into the existing network configuration. Therefore, a seamless transition from the legacy system and setup to an IoT-based configuration is a major challenge. In other words, there is a need to ensure backward compatibility and flexibility in the integration of existing devices.

4) NETWORK TYPE

In the design approach, the network of the IoT healthcare can be one of three essentially different types: data-, service-, and patient-centric architectures. In the data-centric scheme, the healthcare structure can generally be separated into objects based on captured health data. In a service-centric scheme, the healthcare structure is allocated by the assembly of characteristics that they must provide. In the patientcentric scheme, healthcare systems are divided according to the involvement of patients and their family members they consider for treatment. In this regard, determining the appropriate network type for IoT based healthcare solutions becomes an open issue.

5) SCALABILITY

IoT healthcare services, networks, databases, and applications should be scalable because associated operations become more complicated with the addition of various applications as a result of the exponential growth of demands from both individuals and health organizations.

6) CONTINUOUS SURVEILLANCE

There are several cases that require long-term surveillance for the patients such as a patient with a chronic disease. Therefore, providing continuous monitoring for those patients with chronic diseases is very important to make them constantly connected and monitored with healthcare caregivers remotely. In other words, continuous network connectivity is crucial.

7) NEW ILLNESSES AND TURMOIL

The smartphones are being deemed as an interface for IoT healthcare device. Although there are many healthcare apps and new apps are being added to the list every day, the trend has been limited to a few categories of diseases. Research and development activities for new types of diseases and disorders are essential, and the discovery of methods that can make the early detection of rare diseases mobile has long been an important task.

8) IDENTIFICATION

The organizations of the healthcare deal with multi-patient environments generally, wherein many of caregivers perform their responsibilities. In this regard, the proper identification of the caregivers and the patients is needful.

9) DATA PROTECTION

The protection of the captured data from several devices and sensors in healthcare from unauthorized access is critical. Therefore, stringent policies and technical security measures should be introduced to share health data with authorized users, organizations, and applications. An optimal algorithm for collaboration between protection, detection, and reaction services to prevent various attacks, threats, and vulnerabilities is an open challenge.

10) PLATFORMS OF IoT-BASED HEALTHCARE

Hardware architecture of IoT healthcare is more advance than the usual IoT devices since it needs a real-time running system with more rigorous requirements. Therefore, customized computing platforms with run-time libraries are needed. Moreover, for a specific platform, libraries and proper frameworks should be designed so that the developers and designers of the healthcare software can make use of given classes, codes, documents, and other beneficial data more effectively. Hence, a special class of illness-oriented libraries can be helpful.

11) MOBILITY

IoT healthcare network should have the capability to support the movements of patients, where they can be connected anytime and anywhere. The current studies of IoT frameworks for monitoring a patient are still ignoring the movement area and comfort of the patient, where the patient's monitoring is being restricted in terms of place and time.

C. IoT VOICE PATHOLOGY

IoT has brought the sight of a more connected world into reality with a big amount of data and many services that are provided by the heterogeneous networks. On the other hand, cloud computing has protruded and it provides huge storage and gives opportunities for data sharing [52]. The merging of the cloud with the IoT can produce new and many opportunities for both technologies [53]. These technologies may unfold a new horizon of service sharing, the interconnection among the devices, ubiquitous sensing, and provides better cooperation and communication between the people and the things in a more dynamic and distributed performance [54]. Hence, the development of many applications is much needed in several fields of the healthcare sector.

Voice pathology is a quite significant area in the healthcare sector. There are many people who are suffering from voice pathologies due to various reasons such as extreme damage to certain organs, air pollution, smoking, and stress. In a recent study, it has been monitored that more than 7.5 percent of the entire population of the people in the USA suffer from voice pathology [55]. People in certain professions such as teachers and singers are suffering the most from voice turmoil because they utilize their voice exceedingly, where around 20 percent of American teachers have been infected with voice pathologies [55]. The detection or assessment of the voice pathology can be classified into two categories: objective and subjective. The objective assessment does not require specific equipment, where if the algorithm is proper, the outcomes are always unbiased. Meanwhile, the subjective assessment needs specific equipment and trained doctors and thus it incurs a high cost. In addition, the subjective assessment differs from doctor to doctor which relies on the doctor's expertise [56]. On the other hand, the objective assessment can be used only from initial screening, where the final decision must come from the medical doctors.

IV. RELATED WORKS OF IOT IN HEALTHCARE

Recent studies have witnessed a huge interest by researchers and developers in the field of voice pathologies that are based on IoT technology. Table 1 shows a brief summary of some related works for IoT techniques in the healthcare sector. In [57] the authors have proposed a new framework of health surveillance by merging the IoT with the cloud. The paper presented a case study of a voice pathology surveillance in which the signals of the voice are captured through different IoTs and transmitted to a host device. To preserve the authenticity of the voice signals, the signals are watermarked by the patient's identification number. The proposed voice pathology monitoring achieves a very good accuracy with low computation time. The pathological monitoring framework can be enlarged to other kinds of health monitoring utilizing the IoTs and the cloud technologies.

The research paper in [58] presents a framework for smart healthcare surveillance that is connected to the smart city and smart devices for accessible and affordable healthcare. The authors have proposed a Voice Pathology Detection (VPD) approach that has two inputs, electroglottography (EGG) signal and the voice signal. The input devices are connected to the Internet and the obtained signals are sent to the cloud. These signals are then analyzed and classified as either pathological or normal. The results obtained from the signals are passed to the doctors for the ultimate decision to determine the next proper action. The Gaussian mixture model is used as a classification and Saarbruecken Voice Database (SVD) for the database. It has been shown in the paper that the precision of the proposed approach is more than 93 percent.

Meanwhile, the author in [59] has proposed a healthcare surveillance system in a smart home to achieve the needs of elderly people in order to have constant care. The patient's status in this proposed system is monitored via two inputs, speech and video. The microphones and the video cameras are installed in the smart homes; both of these sensors capture the patient's speech and video and send these data to the cloud. The data is processed in the cloud and the classification of the voice signals depend on the patient's case, whether the patient is tensed, normal, or in pain.

The authors in [60] have presented an energy-efficient architecture of the IoT in healthcare applications for scenarios such as home care and clinical. Since the movement of the patients in many statuses is restricted to a room or a building, the proposed architecture is based on the smartly wired gateways that utilize the power from Power over Ethernet (PoE) cables in order to achieve a low-cost and an energyefficient system. Capabilities of PoE allow for powering sensors directly without the need for a different power grid or batteries. Moreover, these gateways connect the wired sensors and hospital devices to web services, which enables the hospital automation and collect the data and the vital signs in a suitable and cost-efficient. Medical sensors are used to measure the patient's status such as heart rate, ECG, temperature, glucose levels, pressure and etc.

The authors of this research paper [61] surveyed the methods and the approaches of the state-of-the-art in the design of efficient and secure healthcare surveillance. In addition, they have proposed an overall framework for the advanced system of healthcare surveillance by describing the complete surveillance life cycle. Also, they have highlighted the primary

TABLE 1.	Summary	of IoT	techniques	in the	healthcare	sector.
----------	---------	--------	------------	--------	------------	---------

Years	Problems	Techniques	Contributions	Limitations	Ref.
2017	Audio pathology surveillance	IoT, Cloud, LBP, and ELM	Realize high efficiency of the detection, the data transmission is secured, and it is easy to use.	Lack of scalability when various inputs merge, such as smart devices and microphones. In addition, there is no framework for dealing with big data of normal and pathological audio in the cloud.	[57]
2017	Detection of the voice pathologies	Cloud, EGG signals, Local Features (LF) as a feature extraction and GMM.	The accuracy of the suggested approach has achieved 93 percent, where it outperformed some state- of-the-art.	In the proposed system, there is no disease diagnosis, where it focuses only on the issue whether the voice is normal or pathological.	[58]
2016	Surveillance of the elderly people and patients	Capturing the status of the elderly or patients in the smart home via video and speech.	The proposed system has achieved an accuracy of 94.68 percent, where it can be effectively utilized in the status surveillance of the patient in the smart home.	The measures of the safety and the security must be taken for ensuring the privacy of patients.	[59]
2015	Energy-consumption used in patient surveillance	IoT healthcare, wired gateways	Created wired gateways which have low-cost and can deploy them along the facilities of the hospital and thus an energy- efficient system in the healthcare.	The wired gateways have been utilized in a small building or room, where the movement was limited.	[60]
2015	Illegally access of patients in the healthcare surveillance system	IoT, WBANs	It presented an overall framework for advanced eHealth surveillance via describing the whole surveillance life cycle. The primary service components were also highlighted.	The authors have not applied solutions for eHealth monitoring in their paper.	[61]
2014	Remote patient surveillance	IoT healthcare, Cloud Computing and Electro Cardiogram (ECG) "Android App"	Building a new application for the Android platform in the healthcare field utilizing the cloud and IoT.	There are no processes in the cloud server in terms of features extraction and classify the signal, it only stores the data.	[62]

service components in the healthcare such as communication networks, servers of medical data processing, and clinic terminals. They have discussed the data collection of patients using WBANs and mobile crowd sensing. Furthermore, this paper has presented and discussed the challenges that need to be considered to develop a secure and efficient surveillance system for patients. Examples of these challenges are usability in terms of interactions between the patient and the system, efficient cost and the quality of the data collection, secure data processing and privacy-preserving.

The authors of this paper [62] presented their work which is an Android application platform in the healthcare field using IoT and cloud. The proposed application is named 'ECG Android App' which provides users with the visualization of their Electro Cardiogram (ECG) waves and also the functionality of data logging. The logged data is uploaded to the centralized cloud of the user's private or a particular medical cloud. The cloud saves all the monitored data and this can be accessed later for analysis by medical doctors.

In [63], the authors have addressed the usages of IoT in the healthcare sector. The chronic disorders prediction in the wearable healthcare devices has been discussed to bring smart healthcare solutions anywhere. In addition, they have discussed several IoT healthcare challenges such as processing enormous volumes of data at a high rapidity, limited network bandwidth, no uniform criteria for data created and hacking connected devices. Retention for a large amount of data could be a difficult task. However, clinically validated instructions and advice have not presented sufficiently.

In [43], an IoT-based smart rehabilitation system was presented. The rehabilitation system is established through Wi-Fi and other technologies such as RFID-based short-distance radio communication technique, Global Positioning System (GPS) technology, and Unique Identifier (UID). The architecture service-oriented is developed and used for designing, implementing, managing and other kinds of healthcare services. After designing and implementation of the IoT rehabilitation system, each and every patient will get good treatment and they are well diagnosed with two rehabilitation strategies, which are resources allocation plan and essential treatment activity information.

In [49], a wireless remote surveillance system for heart rate and oxygen saturation was proposed. In this research paper, the oxygen level in the blood and also the patient's heart rate is measured using the pulse oximeter. Then, this measured data is sent to the central surveillance station via WSN. The patient will be continuously monitored and the central monitoring station receives the information of the patient's oxygen saturation level and heart rate through WSN. If any problem occurs, an alarm will be activated automatically. A Graphical User Interface (GUI) is developed to display the results and measurements of the patients.

An IoT healthcare surveillance system is also introduced in [64]. A prototype (CC2451 sensor tag) is presented for the implementation of the proposed system to gather a patient's data such as humidity, temperature, accelerometer, and pressure. Moreover, they used a chest strap belt as a smart sensor to monitor the heart rate. It alerts about the patient's health condition in real-time, if any problem is experienced, or the patient needs any medical attention and hospitalization. The authors assumed that their work can decrease healthcare cost and increase the specialized care required. However, this work has not presented any development for decreasing the cost, where it is limited by using a present technology that is already available in most patients' homes such as chest strap belt.

Meanwhile, the authors of this paper [65] have worked on implementing fog computing in the healthcare IoT. They have highlighted on the benefit of fog computing, where they have presented a collection of services that make use of the healthcare IoT based on the implementation of an intelligent gateway for fog computing. These services are provided to address the main IoT challenges in the healthcare field such as usability, scalability, performance, and an enormous number of devices which are connected to the Internet that cause many available resources such as computing power and bandwidth.

The study in [66], suggested a system to improve healthcare performance which includes the rural regions by using IoT. The proposed system can monitor Ischemic Heart Disease (IHD) by using a mobile application to upload all patient measurements to the server between the patient and the clinic. These measurements are classified into three stages. Stage I has the patient information such as the sex, weight, height, waist measure, and body mass index. Stage II refers to cholesterol, diabetes, and thyroid. Stage III refers to personal habits, sleeping disturbance, genetic factors and family history. In this system, patients have to report those measurements once a week. Based on the doctors' experience, they will classify the risk of IHD as no danger, low, medium and high dangers. In the case of high danger, ambulance services and doctors have to support the patient rapidly. Patients with low and medium dangers will be informed to visit doctors and take medicines. However, patients' data are collected by patients themselves, where it is very possible for an error to occur when measuring that data and also not all patients have a medical measuring device for measuring cholesterol and diabetes.

V. MACHINE LEARNING

Machine learning was originally proposed as a unique method for Artificial Intelligence (AI) in the late 1950s. It has been gradually developed and has been applied in many applications such as bioinformatics, spam detection, speech recognition and data analysis [67]. Algorithms of machine learning are used as powerful predictors. Since the accuracy of these algorithms is known to improve especially with larger quantities of data to train on, the growing availability of such data in recent years has brought renewed interest to machine learning algorithms [68]. There are two major approaches to machine learning. The first is supervised learning. The supervised learning domain includes the training algorithms utilizing a set of examples. The machine obtains a number of inputs with a specific number of the correct outputs and the learning happens through contrasting empirical results with the correct outputs to recognize the errors [69]. This kind of learning is utilized when past history is used to predict events in the future [70]. The second approach is called unsupervised learning. In this approach, the machine needs to explore the data and try to develop some kind of structure or pattern. The models also should be developed from scratch. This method is often used to determine and differentiate the outliers [71].

VI. MACHINE LEARNING IN HEALTHCARE

Machine learning has a broad societal impact in the healthcare field. In the industry of smartwatches and smart devices that continually collect many health data, the benefit of machine learning in the data analysis is becoming increasingly significant [72]. It can be the resolution for decreasing the high cost of healthcare and also to help establish a better relationship between the patient and the doctor. Machine learning and big data can be applied in many applications of healthcare field; for examples assisting doctors to identify more personalized prescriptions and treatments for patients and also assisting when the patients should schedule appointments for follow up.

Currently, a huge amount of data is available in healthcare. This involves electronic medical records (EMRs) which contain either structured or unstructured data [73]. Structured data in healthcare refers to the information that is easy to classify in a database; they can include a set of categories and statistics such as patient temperatures, patient weights, and also general symptoms such a stomach pain, and headache [74]. However, most of the health data is unstructured data in the form of images, various notes, discharge summaries, reports, videos and audio recording. Data such as personal conversation can refer to numerous various directions [75]. For instance, two patients can have the same exact strain of a cold, but the conversation and data may vary according to the background of the doctor and the patient, and even differ based on various ways of the patient describing the illness symptoms. In general, unstructured data makes up 80 percent of present EMRs and 20 percent structured data.

Since the nature of medicine is related to a kind of narrative, the techniques of modern machine learning should be taken into account when establishing and organizing a relationship among huge amounts of unstructured raw data. The ability of understanding and using this kind of data on a wide scale will be very useful in applying the technologies of machine learning in the healthcare field [76]. For structured data, there are many artificial data technologies currently exist; however, only a few innovators and developers are focusing on the structured data, they only focus on the narratives in the healthcare field. Moreover, when machine learning is applied effectively, it can help doctors to make near-perfect diagnoses, determine the best medicines for patients, and improve the general health of the patients as well as reduce the cost [75].

As the cost of healthcare services stagnates at historically high prices, the used of machine learning in unstructured data can be the solution to this ever-growing issue. In the healthcare sector, 50 percent of the entire costs comes from 5 percent of entire patients; in addition, the number of chronic diseases that require continuous care has progressively increased in the world. Machine learning can recognize patients who may be more likely for frequent diseases. Moreover, close to 90 percent of visits to the emergency room are preventable, where machine learning can be employed to assist in diagnostic and to direct patients for appropriate treatment. Consequently, this reduces costs by keeping patients outside of costly emergency care rooms [75].

A. APPLICATIONS OF MACHINE LEARNING IN HEALTHCARE

The diagnostic reasoning in medical is a very significant area of intelligent systems. The expert systems in these frameworks [77]–[79] present mechanisms for the hypotheses generated from the patient data. For instance, extraction of the rules from the experts' knowledge to build the expert systems. Unfortunately, in numerous statuses, the experts may not be able to formulate which knowledge they should utilize to solve their problems.

Symbolic learning is a technique that can be categorized based on strategies of underlying learning such as learning from discovery, rote learning, learning from examples, and learning by analogy [80]. Techniques of symbolic learning are utilized to enhance learning, and capabilities of the knowledge management for the expert systems [81]. According to the set of clinical statuses, the learning in intelligent systems can be obtained by using the machine learning techniques which is able to produce a systematic explanation for those clinical features. As a result, the knowledge can be expressed in the shape of simple rules such as the KARDIO that has been developed to interpret ECGs.

The authors in [82] present an intelligent system which captures the patient data in real-time during surgery of the cardiac bypass and generates models for the abnormal and normal cardiac physiology to identify changes in patient's status. Moreover, these models can work as initial hypotheses for further experimentation. The learning from patient data faces many difficulties and challenges because of datasets incompleteness (losing of parameters), errors (random noise in the data), and inaccuracy (improper parameters choice for a particular task). Machine learning presents tools and mechanisms for dealing with these properties of medical datasets [83].

Another area of machine learning application in healthcare is the processing of biomedical signal [84]-[86]. Our knowledge of biological systems is incomplete since there are primary information and features hidden in the physiological signals which are not apparent. Furthermore, the effects among the various subsystems cannot be distinguished. Biological signals are distinguished by large variability because of the external stimuli or spontaneous internal mechanisms. Associations among various parameters may become so complicated to be solved with traditional techniques. Machine learning methods and techniques exploit these collections of data to make it simpler, and it can also assist to form relationships which exist among these data and extract features and parameters to improve healthcare. In general, the environment of healthcare depends heavily on computer technology. On the other hand, the application of machine learning methods and techniques can present beneficial aids to help doctors in several cases, tackles the issues related with human fatigue, and facilitates real-time diagnosis and presents fast identification of abnormalities.

B. CHALLENGES OF MACHINE LEARNING IN HEALTHCARE

Notwithstanding the promising results achieved utilizing the machine learning technologies, there are still various unsolved challenges in the clinical healthcare application using machine learning. We highlight the main issues and challenges as follows:

1) DATA SIZE

Machine learning indicates a collection of extremely intensive computational models such as neural networks which are completely connected multi-layer, where a lot of parameters in the network need to be estimated correctly. To obtain this objective, it should provide a huge amount of data. Moreover, understanding illnesses and their variability are more complex than other tasks. For instance, the recognition of the voice pathology type through the speech and or an image processing in the Magnetic Resonance Imaging (MRI) to predict Alzheimer illness. Therefore, from the perspective of the big data, it is crucial to have more amount of medical data to train a robust and effective model in machine learning.

2) THE QUALITY OF DATA

Data in healthcare are extremely heterogeneous, incomplete, and ambiguous. Training a good model in machine learning with such a collection of diversified and huge data is difficult and many issues need to be taken into account such as the sparsity of the data, losing values, and redundancy.

3) TEMPORALITY

The illnesses are constantly changing and progressing across time. However, several proposed models of machine learning in different healthcare fields, assume static inputs of vector-based. These static inputs could not deal with the time factor. A new machine learning method that can deal with temporal medical data is a significant aspect that will need to be developed. In other words, it is vital to design a new approach in machine learning that considers dynamic inputs.

4) FIELD COMPLEXITY

The problems in healthcare applications and biomedicine are becoming more and more complex. The illnesses are highly diverse and there is still no full knowledge of most of the illnesses, how they progress, and their causes. Furthermore, the number of patients in a practical clinical is usually limited, where we cannot ask for more patients as much as we want. Consequently, this leads to a lack of medical training data for an effective machine learning model.

C. MACHINE LEARNING VOICE PATHOLOGY

Machine learning techniques are useful for discriminatory classification processes. These techniques have been utilized in different applications of speech processing, where one of these applications is pathological voice analysis. The classification and the recognition of pathological voice methods are still one of the complicated fields in the research of speech processing. The pathological voice indicates problems in talking that are caused by injuring, mental disease, autism, speech organs abnormality, or other disabilities. The existence of pathologies in the vocal folds affects the natural vibration pattern of the glottis which can cause changes in voice quality. The traditional methods to detect voice pathology are inefficient. They essentially depend on the vocal fold's examination and the specialist study which may generate confused and various evaluations. Furthermore, these traditional methods are costly, consume more time, and need many kinds of equipment [87].

Recent years have witnessed a huge interest in the speech quality evaluation using machine learning techniques due to the importance of this area. The function of this evaluation is to recognize the disorder of the voice and to create a system that is capable of dealing with voice pathologies. The authors in [88] have presented an analysis method to differentiate between pathological and normal voices by applying Gaussian Mixture Model (GMM). GMM is a supervised classification method broadly applied for speaker detection. In addition, they have used Massachusetts Eye and Ear Infirmary (MEEI) database in their work and Mel Frequency Cepstral Coefficient (MFCC) as feature extraction. The authors in [89], also used GMM to distinguish between pathological and normal voices. In their study, normal samples are accurately classified up to 95 percent while 18.3 percent of pathological samples are improperly classified. GMM has been also used for the detection of pathological voices in this study [90], where this method presents a better performance than [89] with an error average reached 7 percent for normal samples 1.4 percent for pathological samples. However, the abovementioned studies have used a limited database for both normal and pathological samples. In addition, there is no improvement in terms of the classification or the feature extraction processes and their studies were limited only in the differentiation of voices to either normal or pathological, where the diagnosis of disease was not included.

VII. RELATED WORKS OF MACHINE LEARNING IN HEALTHCARE

The state-of-the-art technology in healthcare systems has opened up many great opportunities in the implementation of smart environments and smart healthcare. Particularly in the health and medical field, several machine learning algorithms have been developed for different purposes such as to evaluate different types of vital signs speech recognition, voice pathology monitoring, analyses of the voice and diagnosing the disease and etc.

In [91], the authors have presented and developed an Arabic Voice Pathology Database (AVPD) by recording three vowels, isolated words and running speech. Moreover, the perceptual severity for each recorded sample has been provided to being a unique aspect of AVPD. Normal and disordered samples in AVPD were recorded by using the Computerized Speech Lab model 4500 with frequency 48 kHz. All samples were taken in distance of 15 cm between mouth and microphone. Around 51 percent of the total dataset are normal samples (82 female and 116 male) and the remaining are voice disorders samples. The AVPD has included five-voice disorders which are paralysis (14 percent), sulcus (11 percent), vocal fold cysts (7 percent), polyps (11 percent) and nodules (5 percent). In AVPD development, the authors have identified and avoided the shortcomings of various databases of voice disorder. They have used MFCC and Linear Prediction Coefficients (LPC) as speech features extraction and four various machine learning algorithms for the classification process which are GMM, Support Vector Machine (SVM), Vector Quantization (VQ) and Hidden Markov Model (HMM). The MFCC that is used in AVPD is shown in (1), where *m* refers to the corresponding frequency in Mel-scale and f refers to the frequency in Hz. In LPC features extraction, the analysis of Linear Prediction (LP) is applied to reverse filtering on speech signals in order to remove formants effects and thus estimate the source signal. The present sample of the source signal can be estimated through p previous samples as given in (2).

$$m = 2595\log_{10}\left(1 + \frac{f}{700}\right) \tag{1}$$

$$x'_{r} = \sum_{i=1}^{\nu} a_{i} x_{r-i},$$
(2)

where, $x_1, x_2, ..., x_r$ are original speech signal samples and a_i 's refers to the required LPC features. The accuracy of the classification process is calculated by the following equation:

$$Accuracy(\%) = \frac{\text{Total Correctly Detected Samples}}{\text{Total Number of Samples}} + 100$$
(3)

The detection and classification processes results were compared with MEEI database. The highest acquired detection accuracy for AVPD is 81.6 percent and 92.9 percent for the classification, while the accuracy obtained in MEEI is 98.7 percent and 98.2 percent for the detection and classification respectively. Nevertheless, MEEI has overcome AVPD in terms of the classification and detection accuracies, where AVPD has failed in the comparison.

The paper [92] presents a new method named Incremental DBSCAN-SVM for noise detection as well as for voice classification and differentiation. This method utilizes a clustering algorithm called Density-Based Spatial Clustering of Applications with Noise (DBSCAN) and incremental learning for noisy samples detection. DBSCAN is a clustering algorithm that has several advantages, where it can distinguish noises, it works efficiently even in large databases and it can detect clusters of arbitrary shape. Moreover, it used Mahalanobis distance to calculate the distance between the new object and the various means of clusters and also the points that are considered as noises by DBSCAN. The Mahalanobis distance can be calculated as shown in (4) and (5).

$$MD_i = ((x_i - t)^T C_n^{-1} (x_i - t)^{1/2}$$
(4)

where t refers to the multivariate location and C_n refers to the estimated covariance matrix:

$$C_n = \frac{1}{n-1} \sum_{i=1}^n (x_i - \bar{X_n})(x_i - \bar{X_n})^T$$
(5)

The MFCC is used as feature extraction for each voice sample. Next, the output pattern is presented to the SVM classifier to distinguish among pathological and normal voices. This method has the capability of dealing with dynamic and incremental voices database that develops over time. This method has utilized the MEEI database for voice disorders detection, where it has tested 53 normal voices and 173 pathological voices (Adductor, Paralysis, Vocal Polyp and Keratosis). Results based on Incremental DBSCAN-SVM can reach up to 98 percent and outperformed traditional methods. However, the proposed system did not present a specific framework (i.e. IoT, smartphones, cloud) for collecting the voice from patients. Also, Incremental DBSCAN-SVM was evaluated in terms of the accuracy only, where there are other performance measurements that are significant such as G-mean, F-measure, precision, and recall.

The work in [93] proposed a new technique in the detection of voice pathological based on deep neural networks. The idea of this method is to use a recurrent Long Short-Term-Memory (LSTM) layers on a raw audio signal to skip the phase of building the feature vector. The authors obtained voice recordings of the continued vowel /a/ from the SVD database. They used 480 healthy samples and 480 pathological samples, where 70 percent is used for the training, 15 percent for validation and 15 percent for testing the system. Every recording is divided into 64 ms of Hamming windowed and 30 ms overlap. The experimental results have shown that the proposed model achieved 71.36 percent accuracy, 77.67 percent specificity, and 65.04 percent sensitivity based on 206 trained files. Meanwhile, based on 874 tested files, the accuracy of the proposed model is 68.08 percent, the specificity is 77.89 percent, and the sensitivity is 66.75 percent. However, this study needs an enormous amount of data to train the model as well as the accuracy has not achieved a high ratio in the classification process.

The authors in [94] have presented an intelligent solution to analyze voice pathologies by using co-occurrence matrices and GMM. Co-occurrence matrices have been essentially created to extract texture features from images, where these features have proved to be effective in the healthcare sector such as in [95]–[97]. The authors exploited the co-occurrence matrices advantage to design a system of voice pathology analysis. Two kinds of inputs were used: EGG signals and voice signals. The EGG electrodes were used to obtain EGG signals and smartphones were used as sensors to obtain voice signals. Both signals were sent to the cloud, which then were separately processed and combined after the individual classification phase. The experiments were evaluated based on the SVD database [98]. The authors analyzed around 400 samples that uttered vowel sound /a/. Pathological samples included in this method are vocal fold paralysis, sulcus, vocal fold cyst, vocal fold polyp and vocal fold nodules. The co-occurrence matrices are applied on the image in two distances (d1 and d2) and two directions($\emptyset = 0^{\circ}, 90^{\circ}$). The intention of d1 is to compare between two immediate neighbourhoods in terms of the energy levels, and d2 is to compare with the next nearest neighborhoods. The cooccurrence matrices are calculated based on 12 (frame) \times 24 (filter) window which can be calculated as follows:

$$COOCCUR_{x,y} = \sum_{u=1}^{A} \sum_{v=1}^{B} (\mathrm{Im}_{u,v} = x) \bigwedge (\mathrm{Im}_{u',v'} = y) \quad (6)$$

where, Im refers to image size (A × B), x and y refer to filters energy levels, u' and v' refer to distances (d1 and d2) and directions (0°, 90°). The accuracy of combined EGG and voice signals achieved 99.98 percent which is higher than the compared studies in [98]–[100]. However, there is no details explanation on the dataset used, where the number of normal samples and pathological samples used in the experiment is unknown. Moreover, the distance between the speaker and the EGG electrodes or smartphone is not mentioned.

The parameters of the MFCC (e.g. filter space, frame size, window length and filter bandwidths) are also considered in many other studies, such as in [101], [102]. In [101], the selected patients are suffering from spasmodic dysphonia. The authors have combined SVM and GMM classifiers and used RBF kernel (Radial Basis Function) to differentiate normal and pathological voices. Moreover, they have modified and used Bhattacharyya (Bh) and Kullback-Leibler (KL) distances for measuring the distance between GMMs in order to enhance the capacities of GMM discriminative. The obtained results have shown that the system has 2 percent and 4 percent of the sensitivity improvement when KL and Bh distances were utilized. However, the performance of SVM and GMM

algorithms with MFCC as signal features is tested only on a limited dataset that is obtained from MEEI database. The work in [102] has presented a system to evaluate the voice signal in terms of voice pathology diagnosis. This system has three classes; the first one is for 36 samples of normal voices, and the second class for pathological voices, where it includes 19 samples of vocal fold nodules and 40 edemas samples. The third class is for the neuromuscular disorder that includes 59 vocal fold unilateral paralysis samples. It used two classes of voice pathology to verify which acoustic feature is more reliable and has more pathology information. The voices are obtained from MEEI database with the sustained vowel /a/ and it used GMM for the classification process. However, the obtained results have shown that the accuracy only reached 77.90 percent.

The SVM method is also applied in [103] to predict the existence of dysphonia and to study four kinds of pathology: spasmodic dysphonia, cysts, chronic laryngitis, and Reinke's edema. It has used MFCC as feature extraction and SVM as a classifier. The authors proposed a method that is based on Linear Discriminant Analysis (LDA) as a reduction method of MFCC dimensionality. The purpose of LDA is to estimate matrix parameters to determine features from h-dimensional to k-dimensional (k<h) and thus the matrix within the class will be minimized. This matrix is computed as below equation:

$$S_w = \frac{1}{N} \sum_{i=1}^k N_i \Sigma_i \tag{7}$$

where, N refers to the number of MFCC features. SVM method classified the pathological voice signal with an accuracy that is up to 86 percent. However, this method was only tested on a limited dataset. Indeed, only 40 healthy and 70 pathological voices were selected through SVD database.

The authors in [104] have proposed an analysis of the speech signal by applying the GMM classifier, MFCC extractor with various jitter and shimmer for the detection of the neurological disorder voice. Equations (8) and (9) are used to calculate Jitter and Shimmer in order to evaluate the percentage of the speech signal.

$$Jitter = \frac{\frac{1}{N-1} \sum_{k=1}^{N} |T_k - T_{k+1}|}{\frac{1}{N} \sum_{k=1}^{N} |T_k|}$$
(8)

Shimer =
$$\frac{\frac{1}{N-1}\sum_{k=1}^{N}|A_k - A_{k+1}|}{\frac{1}{N}\sum_{k=1}^{N}|A_k|}$$
 (9)

 T_k refers to time, A_k indicates an amplitude, and N refers to number of cycles. For normal and pathological voices, authors used SVD dataset with vowel sounds /u/ and /a/, where they have chosen 52 normal samples and 29 pathological samples. However, this study was evaluated on a small set of SVD database involving only 52 healthy voices and 29 pathological voices.

Different classification models are explored and compared in [105] to obtain the capability of vocal parameters in distinguishing pathological voices from normal voices. The aim of this work is to distinguish pathological and normal voices of children by using various classification techniques such as SVM and Radial Basis Functional Neural Network (RBFNN). The dataset is created by recording voices from 20 children, in which 10 samples are pathological voices and 10 samples are normal voices). The total of samples is 8000 per second. These voices are used to train and test the classifiers. Next, the signal of the speech is analyzed to extract the vocal parameters such as frequencies, and signal energy. Results have shown that RBFNN has an accuracy that reached 91 percent, while the SVM achieved 83 percent accuracy. However, this study was performed on a limited dataset that involves children voices only, where the dataset should be varied and has different samples of the voice such as gender, age and different voice pathologies.

The research paper in [106] aims to extend the traditional Principal Component Analysis (PCA) by utilizing dynamic representation in the classification task. PCA is a method that is widely used as a feature extraction, where it can correctly select a relevant subset of original features. The method of dynamic features with PCA is proposed in order to reduce the feature space dimension which causes the complexity of the classifier and the computational cost. In this method, three classifiers have been used which are Bayes, SVM and GMM. These classifiers are applied in three scenarios to evaluate the performance of proposed dynamic features that used PCA method. In the first scenario, classifiers are applied without using any feature extraction. In the second scenario, the classifiers are applied by using feature extraction. Meanwhile, classifiers are used with the proposed dynamic feature extraction and PCA in the third scenario. This method is applied for pathology voice detection in the speech and it is evaluated based on two different databases of voice disorders, which are MEEI and Universidad Politécnica de Madrid (UPM). It used 53 of normal voices and 173 of the pathological voices. The dataset is divided into 70 percent for the training phase and 30 percent for testing. The recorded voices are framed with 40 ms and the dynamic feature consists of 32time vectors which represent each whole voice sample. In the MEEI database, the accuracy rate can reach 95 percent and the space dimension decreased from 32 to 4 variables. On the other hand, in UPM database, the dimensionality decreased from 32 to 9 and achieved 80 percent of the accuracy rate. Furthermore, it is possible in the proposed system to identify the original dynamic features that are pertinent to the task of detecting pathological voice. However, the proposed method has failed in terms of sensitivity.

The SVM classifier is also used in [107] to differentiate between pathological voices and normal voices. The aim of this study is to reduce the recognition error rate by applying MFCC with Fast Fourier Transform (FFT). FFT-based MFCC parameters are obtained by calculating DCT (Discrete Cosine Transform) over the logarithm of the energy in different

TABLE 2. Summary of machine learning algorithms used in the pathological voices systems.

Years	Problems	Classifiers	Features Extractions	Databases	Pathological Samples	Normal Samples	Accuracy	Ref.
2020	Pathological voice detection	GEV, GMM and GNB	H-KLD, HASS-KLD and MFCC	MEEI	173	53	99.55%	[111]
2020	Voice pathologies detection and classification	SVM	HOS, EMD and DWT	SVD and Private- Tunis	SVD = 569 Private = 28	SVD = 561 Private = 30	SVD = 99.26% detection, 100% classification. Private = 94.82% detection, 94.44% classification	[112]
2020	Speech pathologies investigation	DPM	ZCR, SE and SH	SVD	126	687	95%	[113]
2020	Developmental dysphasia	SVM and FFNN	Glottal features, MFCC and openSMILE	SLI	54	44	SVM = 88% - 96%, and FFNN = 95% - 99%	[114]
2020	Parkinson disease detection	k-NN, MLP, SVM, and RF	Baseline features, MFCC, DWT and TQWT	Private- Turkey	564	193	94.7%	[115]
2019	Voice pathology surveillance	CNN	Parallel CNNs	SVD	1342	686	95.5%	[116]
2019	Dysphonia disease	CNN	STFT, FSST and FSST+DA	SVD	94	94	70%	[117]
2019	Voice pathology analysis	k-NN and SVM	DT-CWPT	MEEI and SVD	MEEI = 106 SVD = 244	MEEI = 201 SVD = 592	SVM (MEEI- SVD) = 94% - 97.69%. k-NN (MEEI-SVD) = 80% - 87%	[118]
2019	Voice disorders detection	SVM and k-NN	Glottal signal parameters	SVD and Private- ISR	71	34	SVM = 98.5% and k-NN = 88.2 %	[119]
2019	Voice pathology detection	SVM	CNN	TUH	998	1385	87.32%	[120]
2019	Detection of pathological voice	SVM	Glottal source features and MFCC	HUPA and SVD	HUPA = 200 SVD = 857	HUPA = 239 SVD = 661	HUPA = 78.37% and SVD = 74.32%	[121]
2018	Voice pathology detection	SVM	CNN	MEEI and SVD	MEEI = 95 SVD = 244	MEEI = 53 SVD = 262	MEEI = 73.3% SVD = 98.77%	[122]
2018	Voice pathology detection and classification	SVM	Frequency bands	MEEI, SVD and AVPD	MEEI = 101 SVD = 263 AVPD = 127	MEEI = 53 SVD = 266 AVPD = 169	MEEI = 99.54% SVD = 99.53% AVPD = 96%	[123]
2018	Voice disorder identification	SVM	HNR, Jitter, Shimmer and MFCC	SVD	685	685	85.77%	[124]
2018	Voice pathology detection	ANN and SVM	Glottal flow parameters	SVD and MEEI	SVD = 251 MEEI = 180	SVD = 260 MEEI = 53	SVD = 99.27% MEEI = 93.66%	[125]
2017	Arabic voice pathology surveillance system	SVM, VQ, GMM, and HMM	MFCC, LPC, LPCC, PLP, RASTAPLP, and MDVP	AVPD and MEEI	173	53	AVPD = 81.6% MEEI = 98.7%	[91]
2017	The distinguish	SVM.	MFCC	MEEI	173	53	98.63%	[92]

TABLE 2. (Continued.) Summary of machine learning algorithms used in the pathological voices systems.

	among pathological and healthy voices							
2017	The voice pathology detection	DNN	LSTM	SVD	1356	687	71.36%	[93]
2017	The differentiating pathological voices from normal voices	GMM	Co- occurrence matrices	SVD	1320	650	99.87%	[94]
2016	Pathological voices detection	SVM	Pitch and LPC	SVD	160	80	86%	[126]
2016	Voice pathology assessment for big data	SVM GMM and ELM	MPEG-7 and IDP features	SVD and MEEI	SVD = 1800 MEEI = 600	SVD = 1800 MEEI = 53	SVM = 73.2% GMM = 70.4% ELM = 80.4%	[127]
2016	Voice disorders identification	ANN	LDA and MFCC	SVD	70	50	87.82	[128]
2016	Detection of voice disorders	SVM and ANN	MMTLS	MEEI and Private	MEEI = 657 Private = 527	MEEI = 53 Private = 6	SVM = 94.51% ANN = 96.48 %	[129]
2016	Voice pathology detection	SVM and GMM	MFCC	SVD	40	60	96.5% - 95.5%	[101]
2015	The voice pathology identification system	GMM	MFCC	MEEI	118	36	77.90%	[102]
2015	The detection of pathological and normal samples of the voice	SVM	MFCC	SVD	70	50	86%	[103]
2014	The detection of the pathological voices in the spasmodic dysphonia	GMM	MFCC, jitter and shimmer	SVD	38	63	82.37%	[104]
2014	The distinguish among pathological and normal voices	SVM and RBFNN	Acoustic feature extraction	Private	10	10	83% for SVM and 91% for RBFNN	[105]
2009	The detection of the pathology existence in the speech	Bayes Classifier, SVM, and GMM	Dynamic feature extraction	MEEIVL and UPM	173	53	95% for MEEIVL and 80% for UPM	[106]
2005	The detection of the pathology existence from voice records	SVM	MFCC	MEEI	77	53	95%	[107]
2002	The automatic detection of normal and pathologies voices	НММ	MFCC	MEEI	662	53	99.44%	[108]

frequencies as shown in the following equation:

$$C_m = \sum_{k=1}^{M} \log(S_k) \cos\left[m(k-0.5)\frac{\pi}{M}\right]$$
(10)

where, $1 \le m \le L$ (*L* is the analysis order), S_k refers to several frequency bands and *M* is the band number in Mel scale. The features are calculated from short-time windows that are extracted from the utterances, where the window length is set to contain two consecutive pitch periods. In addition, feature extraction is applied using 40 ms and the rate of frames gained is 50 frames. Hamming windowing is selected with an overlap 50 percent among adjacent frames. The decision of the voice signal classification whether normal or pathological is based on the threshold over the number of frames. The threshold is based on 80 percent of the frame error, where if 80 percent frames are classified as normal voice, then the final decision is normal class, otherwise it will be classified as pathological class. Furthermore, the model has used MEEI database, where it has selected 53 normal samples and 77 pathological samples. In addition, 70 percent of the samples are used for training and 30 percent is for validation. The authors have concluded that FFT-based MFCC performed faster than Multilayer Perceptron (MLP). Moreover, SVM performance with FFT-based MFCC achieved 95 percent classification accuracy and 5 percent error rate.

With regards to the voice pathology recognition, HMM classifier is considered as a competitor to the other powerful classifiers. HMM is used to distinguish among pathological voices and normal voices in [108]. This study used 710 voice samples with 53 normal samples and 657 pathological samples. All samples are obtained from MEEI database. The authors focused on the sustained vowel /a/ in voice samples. Along with those samples, the dataset has also acoustic speech sample that is called "Rainbow passage" which has 662 pathological subjects that belong to the same patients who have provided the vowel /a/ samples. Also, for the normal class, there are 53 recordings with a length 12 sec of Rainbow passage. In addition, MFCC was modelled by GMM in HMM classifier. To compare and evaluate the proposed method results, there is an improvement factor that can be defined as follows:

$$improvement = \frac{Accuracy}{Base_Accuracy} - 1$$
(11)

where, *Base_Accuracy* refers to the rate of the correct classification of HMM classifier for the vowel /a/, and *Accuracy* refers to the rate of the correct classification for the data that has been processed. This study has outperformed other methods such as a Neural Network (NN) and Nearest Mean Classifier (NMC), where the classification accuracy rate for HMM is 99.40 percent and 98.59 percent for the Rainbow passage. However, the pathological voice samples are divided unequally, where there are many samples for a certain disease such as Hyper-function and also there are very few samples for another disease such as adductor spasmodic dysphonia. Consequently, this affects the classification accuracy rate.

From the recent studies in the above mentioned literature, many systems have been proposed for voice pathology detection [98]–[100], [109], [110]. However, these systems aimed to differentiate normal voices from pathological voices only, where there are research shortages in terms of voice disease diagnosis with respect to laryngeal cancer. In machine learning algorithms, there are cases where voice signals could not guarantee high accuracy and cause time-consumption in the pathology monitoring methods. Consequently, there is an urgent need for a study and highlights the most important issues and challenges facing voice pathology systems and the significance of the disease diagnosis in voice pathology. Furthermore, in terms of machine learning algorithms, we highly recommend choosing the best feature extraction method that can handle the audio pathology according to the dimensions of the feature extraction. Moreover, the machine learning classifier should be developed by either proposing a new activation function or tuning on a mathematical function of that classifier scheme. Thus, it can improve the performance of a system in terms of effectiveness and efficiency. Table 2 shows a brief summary of the related works for machine learning algorithms used in the pathological voices systems.

VIII. CONCLUSION

The technologies of IoT, cloud and machine learning algorithms are used in many aspects of the healthcare sector. These technologies will help to develop the healthcare applications in general and in the voice pathology surveillance methods in particular.

In this paper, we have provided an inclusive overview of the state of the art techniques which used IoT and machine learning methods in several healthcare fields, particularly in the voice pathology assessment systems. Moreover, we have highlighted and discussed the main related research results, issues, applications and recent studies which were implemented in the healthcare field and in particular in the voice pathology.

Based on this survey, we can summarize our review as follows:

- IoT is often not completely and correctly exploited in the healthcare area because of the excessively big interval between the data collection and the capability to process and analyze.
- The main problem that faces each patient, especially living in remote locations is the unavailability of doctors and treatment in critical cases. Thus, there is a need for providing a new framework by using IoT devices in those areas.
- There are many situations in which patients require longterm surveillance (e.g., a patient with a chronic disease). In this regard, the provision of constant monitoring is very important.
- Research and Development (R&D) activities for a new type of voice diseases and disorders are essential, and the

discovery of methods that can make the early detection of rare diseases mobile has long been an important task.

- The IoT healthcare network must have the ability to support the mobility of patients such that they can be connected anywhere. This would require a comprehensive IoT ecosystem which considers customization of connectivity, without compromising on the data security.
- Most of the researchers are focusing on the voice pathology and their finding is only limited to differentiating either the voice is normal (healthy) or pathological voice, where there is a need for a complete IoT framework for a specific disease detection such as laryngeal cancer. Since laryngeal cancer is life threatening illness, a new and effective framework/model for laryngeal cancer early detection is much needed.
- Some patients with sound disorder that could potentially be caused by laryngeal cancer, may not want to do further medical examination because they thought that the hoarseness in their voice is not important or dangerous in their life (caused by other diseases such as flu or inflammation of the larynx), or perhaps they do not have time to go there and undergo several processes. Therefore, this is a significant issue which requires a seamless framework that provides an effective and noninvasive method for the diagnosis of chronic diseases such as laryngeal cancer.
- Since machine learning algorithms are used in medical diagnosis, the accuracy of these algorithms must be high enough to avoid any errors in the classification process. Moreover, the developed algorithms must take into account data protection requirements, as well as ethical and regulatory issues in order to merit the trust of patients and healthcare providers and prevents unnecessary risks.
- The slow acceptance for applying the machine learning technology in the medical diagnosis, where this technology should obtain more support from organizations and integrated with the current instrumentation to present simple use for both doctors and patients.
- There are many features extractions and classifiers which are used in different domains such as image processing and text identification. In healthcare, the situation is so critical where one must choose the best feature extraction and classifier that are used accurately and effectively in the voice pathology detection.
- The lack of providing a specific dataset for a particular disease such as laryngeal cancer, where the existing dataset contains different voice diseases. In addition, the distribution of voice pathology samples is very unequal which makes the detection of voice pathology a difficult problem. The reason is due to a particular type of voice pathology which occurred only once in the whole dataset such as SVD. Thus, that type of voice pathology could not be trained well and resulting in low accuracy.

The above conclusions based on the conducted survey in this article must be carefully highlighted in future healthcare studies. Therefore, IoT frameworks and machine learning algorithms in such a field need to have high effectiveness and efficiency in order to present a simple process for patients to check up their health easily by IoT devices and also to obtain a high diagnostic accuracy of a particular disease by adopting machine learning classifiers.

ACKNOWLEDGMENT

The authors would like to thank the Ministry of Higher Education (MOHE), Research Management Centre of Universiti Teknologi Malaysia (UTM) and School of Electrical Engineering, UTM, for the sponsorship and advice for this research.

REFERENCES

- Z. Pang, "Technologies and architectures of the Internet-of-Things (IoT) for Health and Well-being," Ph.D. dissertation, Dept. Electron. Syst., KTH Royal Inst. Technol., Stockholm, Sweden, Jan. 2013.
- [2] S. B. Baker, W. Xiang, and I. Atkinson, "Internet of Things for smart healthcare: Technologies, challenges, and opportunities," *IEEE Access*, vol. 5, pp. 26521–26544, 2017.
- [3] K. Vasanth and J. Sbert, "Creating solutions for health through technology innovation," *Texas Instrum.*, vol. 1, pp. 1–5, Oct. 2016.
- [4] S. M. Riazul Islam, D. Kwak, M. Humaun Kabir, M. Hossain, and K.-S. Kwak, "The Internet of Things for health care: A comprehensive survey," *IEEE Access*, vol. 3, pp. 678–708, 2015.
- [5] M. Fatima and M. Pasha, "Survey of machine learning algorithms for disease diagnostic," J. Intell. Learn. Syst. Appl., vol. 9, no. 1, pp. 1–16, 2017.
- [6] B. Erickson, P. Korfiatis, Z. Akkus, and T. Kline, "Machine learning for medical imaging," *RadioGraphics*, vol. 37, no. 2, pp. 505–515, 2017.
- [7] G. D. Magoulas and A. Prentza, "Machine learning in medical applications," in Advanced Course on Artificial Intelligence. Berlin, Germany: Springer, Sep. 2001, pp. 300–307.
- [8] C. Thota, R. Sundarasekar, G. Manogaran, R. Varatharajan, and M. Priyan, "Centralized fog computing security platform for IoT and cloud in healthcare system," in *Exploring Convergence Big Data Internet Things*. Philadelphia, PA, USA: IGI Global, 2018, pp. 141–154.
- [9] C. Doukas and I. Maglogiannis, "Bringing IoT and cloud computing towards pervasive healthcare," in *Proc. 6th Int. Conf. Innov. Mobile Internet Services Ubiquitous Comput.*, Palermo, Italy, Jul. 2012, pp. 922–926.
- [10] L. Salhi, T. Mourad, and A. Cherif, "Voice disorders identification using multilayer neural network," *The Int. Arab J. Inf. Technol.*, vol. 7, no. 2, p. 8, Apr. 2010.
- [11] M. Hariharan, M. P. Paulraj, and S. Yaacob, "Time-domain features and probabilistic neural network for the detection of vocal fold pathology," *Malaysian J. Comput. Sci.*, vol. 23, no. 1, pp. 60–67, 2010.
- [12] C. E. Martinez and H. L. Rufiner, "Acoustic analysis of speech for detection of laryngeal pathologies," in *Proc. 22nd Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, Jul. 2000, pp. 2369–2372.
- [13] D. Bone, C.-C. Lee, T. Chaspari, J. Gibson, and S. Narayanan, "Signal processing and machine learning for mental health research and clinical applications [Perspectives]," *IEEE Signal Process. Mag.*, vol. 34, no. 5, pp. 195–196, Sep. 2017.
- [14] Y. Hoon Joo, J.-K. Cho, S.-H. Ahn, H. J. Hong, S. Y. Kwon, and K. H. Kwon, "Guidelines for the surgical management of laryngeal cancer: Korean Society of Thyroid-Head and Neck Surgery," *Clin. Experim. Otorhinolaryngology*, vol. 10, p. 1, Feb. 2019.
- [15] A. W. Anwer, M. Faisal, A. A. Malik, A. Jamshed, R. Hussain, and M. T. Pirzada, "Head and neck cancer in a developing country-a hospital based retrospective study across 10 years from Pakistan," *J. Cancer Allied Specialties*, vol. 3, no. 4, p. S104, 2018.
- [16] T. A. D. Souza, V. J. Vieira, M. A. D. Souza, S. E. Correia, S. C. Costa, and W. C. D. A. Costa, "Feature selection based on binary particle swarm optimisation and neural networks for pathological voice detection," in *Proc. Latin America Congr. Comput. Intell. (LA-CCI)*, Curitiba, Brazil, Oct. 2015, pp. 1–6.

- [17] T. Lee, Y. Liu, P.-W. Huang, J.-T. Chien, W. K. Lam, Y. T. Yeung, T. K. T. Law, K. Y. S. Lee, A. P.-H. Kong, and S.-P. Law, "Automatic speech recognition for acoustical analysis and assessment of cantonese pathological voice and speech," in Proc. IEEE Int. Conf. Acoust., Speech Signal Process. (ICASSP), Shanghai, China, Mar. 2016, pp. 6475-6479.
- [18] J.-A. Gómez-García, L. Moro-Velázquez, J. I. Godino-Llorente, and G. Castellanos-Domínguez, "Automatic age detection in normal and pathological voice," in Proc. 16th Annu. Conf. Int. Speech Commun. Assoc., Dresden, Germany 2015, pp. 3739-3743.
- [19] G. Muhammad and M. Melhem, "Pathological voice detection and binary classification using MPEG-7 audio features," Biomed. Signal Process. Control, vol. 11, pp. 1-9, May 2014.
- [20] X. Cui, "The Internet of Things," in Ethical Ripples of Creativity and Innovation. London, U.K.: Springer, 2016, pp. 61-68.
- [21] X.-Y. Chen and Z.-G. Jin, "Research on key technology and applications
- for Internet of Things," *Phys. Procedia*, vol. 33, pp. 561–566, Jun. 2012. [22] K. Ashton, "That 'Internet of Things' thing," *RFID J.*, vol. 22, pp. 97-114, Jun. 2009.
- [23] F. Liu, H. Ning, H. Yang, Z. Xu, and Y. Cong, "RFID-based EPC system and information services in intelligent transportation system," in Proc. 6th Int. Conf. Telecommun., Chengdu, China, Jun. 2006, pp. 26-28.
- [24] M. Zhang, F. Sun, and X. Cheng, "Architecture of Internet of Things and its key technology integration based-on RFID," in Proc. 5th Int. Symp. Comput. Intell. Design, Hangzhou, China, Oct. 2012, pp. 294-297.
- [25] F. E. Idachaba and U. Tommy, "RFID laptop monitoring and management system," in Proc. World Congr. Eng., London, U.K. Jul. 2014, pp. 1-4.
- [26] H. Ning and H. Liu, "Cyber-Physical-Social based security architecture for future Internet of Things," Adv. Internet Things, vol. 2, no. 1, pp. 1-7, 2012.
- [27] J. Gubbi, R. Buyya, S. Marusic, and M. Palaniswami, "Internet of Things (IoT): A vision, architectural elements, and future directions," Future Gener. Comput. Syst., vol. 29, no. 7, pp. 1645-1660, Sep. 2013.
- [28] L. Butgereit, L. Coetzee, and A. C. Smith, "Turn me on! using the 'Internet of Things' to turn things on and off," in Proc. 6th Int. Conf. Pervasive Comput. Appl., Port Elizabeth, South Africa, Oct. 2011, pp. 4-10.
- [29] G. R. González, M. M. Organero, and C. D. Kloos, "Early infrastructure of an Internet of Things in spaces for learning," in Proc. 8th IEEE Int. Conf. Adv. Learn. Technol., Santander, Cantabria, Spain, 2008, pp. 381-383.
- [30] Y. Huang and G. Li, "Descriptive models for Internet of Things," in Proc. Int. Conf. Intell. Control Inf. Process., Dalian, China, Aug. 2010, pp. 483-486.
- [31] O. Vermesan, P. Friess, P. Guillemin, S. Gusmeroli, H. Sundmaeker, A. Bassi, I. S. Jubert, M. Mazura, M. Harrison, M. Eisenhauer, and P. Doody, "Internet of Things strategic research roadmap," in Internet of Things: Global Technological and Societal Trends, vol. 1, O. Vermesan, P. Friess, P. Guillemin, S. Gusmeroli, H. Sundmaeker, and A. Bassi, Eds. Gistrup, Denmark: River Publishers, 2011, pp. 9-52.
- [32] A. Bujari and C. E. Palazzi, "Opportunistic communication for the Internet of everything," in Proc. IEEE 11th Consum. Commun. Netw. Conf. (CCNC), Las Vegas, NV, USA, Jan. 2014, pp. 502-507.
- [33] J. Miranda, N. Makitalo, J. Garcia-Alonso, J. Berrocal, T. Mikkonen, C. Canal, and J. M. Murillo, "From the Internet of Things to the Internet of people," IEEE Internet Comput., vol. 19, no. 2, pp. 40-47, Mar. 2015.
- [34] I. Kerr, "The internet of people? Reflections on the future regulation of human-implantable radio frequency identification," in Privacy, Identity, and Anonymity: Lessons From the Identity Trail, I. Kerr, V. Steeves, and C. Lucock, Eds. Oxford, U.K.: Oxford Univ. Press, Feb. 2013, ch. 19, pp. 335-357. [Online]. Available: https://ssrn.com/abstract=2225565
- [35] D. E. O'Leary, "Big Data', The 'Internet of Things' and the 'Internet of signs," Intell. Syst. Accounting, Finance Manage., vol. 20, pp. 53-65, Mar. 2013.
- [36] M. Nitti, V. Pilloni, G. Colistra, and L. Atzori, "The virtual object as a major element of the Internet of Things: A survey," IEEE Commun. Surveys Tuts., vol. 18, no. 2, pp. 1228-1240, 2nd Quart., 2016.
- [37] C. E. Turcu and C. O. Turcu, "Internet of Things as key enabler for sustainable healthcare delivery," Procedia Social Behav. Sci., vol. 73, pp. 251-256, Feb. 2013.
- [38] S. Tyagi, A. Agarwal, and P. Maheshwari, "A conceptual framework for IoT-based healthcare system using cloud computing," in Proc. 6th Int. Conf. Cloud Syst. Big Data Eng. (Confluence), Noida, India, Jan. 2016, pp. 503-507.

- [39] A. Kulkarni and S. Sathe, "Healthcare applications of the Internet of Things: A review," Int. J. Comput. Sci. Inf. Technol., vol. 5, pp. 6229-6232, May 2014.
- [40] S. Neelam, "Internet of Things in Healthcare," M.S. thesis, Dept. Creative Technol., Blekinge Inst. Technol., Karlskrona, Sweden, 2017.
- [41] M. B. Blake, "An Internet of Things for healthcare," IEEE Internet Comput., vol. 19, no. 4, pp. 4-6, Jul. 2015, doi: 10.1109/MIC.2015.89.
- [42] A. A. Roman Richard, M. F. Sadman, U. H. Mim, I. Rahman, and M. S. R. Zishan, "Health monitoring system for elderly and disabled people," in Proc. Int. Conf. Robot., Elect. Signal Process. Techn. (ICREST), Dhaka, Bangladesh, Jan. 2019, pp. 677-681.
- [43] G. Yang, J. Deng, G. Pang, H. Zhang, J. Li, B. Deng, Z. Pang, J. Xu, M. Jiang, P. Liljeberg, H. Xie, and H. Yang, "An IoT-enabled stroke rehabilitation system based on smart wearable armband and machine learning," IEEE J. Transl. Eng. Health Med., vol. 6, 2018, Art. no. 2100510.
- [44] K. Natarajan, B. Prasath, and P. Kokila, "Smart health care system using Internet of Things," J. Netw. Commun. Emerg. Technol. (JNCET), vol. 6, pp. 37-42, Mar. 2016.
- [45] L. Yang, Y. Ge, W. Li, W. Rao, and W. Shen, "A home mobile healthcare system for wheelchair users," in Proc. IEEE 18th Int. Conf. Comput. Supported Cooperat. Work Design (CSCWD), Hsinchu, Taiwan, May 2014, pp. 609-614.
- [46] C. L. Ventola, "Mobile devices and apps for health care professionals: Uses and benefits," Pharmacy Therapeutics, vol. 39, no. 5, pp. 356-364, May 2014.
- [47] Y. Jie Fan, Y. Hong Yin, L. Da Xu, Y. Zeng, and F. Wu, "IoT-based smart rehabilitation system," IEEE Trans. Ind. Informat., vol. 10, no. 2, pp. 1568-1577, May 2014.
- [48] B. Tan and O. Tian, "Short paper: Using BSN for tele-health application in upper limb rehabilitation," in Proc. IEEE World Forum Internet Things (WF-IoT), Seoul, South Korea, Mar. 2014, pp. 169-170.
- [49] Z. Guangnan and L. Penghui, "IoT (Internet of Things) control system facing rehabilitation training of hemiplegic patients," Chin. Pat., vol. 202, no. 184, p. 661, 2012.
- [50] Y. Yue-Hong, F. Wu, F. Y. Jie, L. Jian, X. Chao, and Z. Yi, "Remote medical rehabilitation system in smart city," Chin. Pat., vol. 103, p. 880, Jan. 2014.
- [51] C. Rotariu and V. Manta, "Wireless system for remote monitoring of oxygen saturation and heart rate," in Proc. Federated Conf. Comput. Sci. Inf. Syst. (FedCSIS), Wroclaw, Poland, Sep. 2012, pp. 193-196.
- [52] H. Atlam, R. Walters, and G. Wills, "Fog computing and the Internet of Things: A review," Big Data Cognit. Comput., vol. 2, no. 2, p. 10, 2018.
- X. Li, Q. Wang, X. Lan, X. Chen, N. Zhang, and D. Chen, "Enhanc-[53] ing cloud-based IoT security through trustworthy cloud service: An integration of security and reputation approach," IEEE Access, vol. 7, pp. 9368-9383, 2019.
- [54] A. Botta, W. de Donato, V. Persico, and A. Pescapé, "Integration of cloud computing and Internet of Things: A survey," Future Gener. Comput. Syst., vol. 56, pp. 684-700, Mar. 2016.
- [55] N. Bhattacharyya, "The prevalence of voice problems among adults in the united states," Laryngoscope, vol. 124, no. 10, pp. 2359-2362, Oct. 2014.
- [56] G. Muhammad, M. F. Alhamid, M. Alsulaiman, and B. Gupta, "Edge computing with cloud for voice disorder assessment and treatment," IEEE Commun. Mag., vol. 56, no. 4, pp. 60-65, Apr. 2018.
- G. Muhammad, S. M. M. Rahman, A. Alelaiwi, and A. Alamri, "Smart health solution integrating IoT and cloud: A case study of voice pathology monitoring," IEEE Commun. Mag., vol. 55, no. 1, pp. 69-73, Jan. 2017.
- [58] M. S. Hossain, G. Muhammad, and A. Alamri, "Smart healthcare monitoring: A voice pathology detection paradigm for smart cities," Multimedia Syst., vol. 25, no. 5, pp. 565-575, Oct. 2019.
- [59] M. S. Hossain, "Patient status monitoring for smart home healthcare," in Proc. IEEE Int. Conf. Multimedia Expo Workshops (ICMEW), Seattle, WA, USA, Jul. 2016, pp. 1-6.
- [60] J. Granados, A.-M. Rahmani, P. Nikander, P. Liljeberg, and H. Tenhunen, "Web-enabled intelligent gateways for eHealth Internet-of-Things," in Internet of Things. User-Centric IoT. Cham, Switzerland: Springer, 2015, pp. 248-254.
- [61] A. Sawand, S. Djahel, Z. Zhang, and F. Nait-Abdesselam, "Toward energy-efficient and trustworthy eHealth monitoring system," China Commun., vol. 12, no. 1, pp. 46-65, Jan. 2015.
- [62] J. Mohammed, C.-H. Lung, A. Ocneanu, A. Thakral, C. Jones, and A. Adler, "Internet of Things: Remote patient monitoring using Web services and cloud computing," in Proc. IEEE Int. Conf. Internet Things (iThings), IEEE Green Comput. Commun. (GreenCom) IEEE Cyber, Phys. Social Comput. (CPSCom), Taipei, Taiwan, Sep. 2014, pp. 256-263.

- [63] D. K R and A. K R, "A comprehensive review on usage of Internet of Things (IoT) in healthcare system," in *Proc. Int. Conf. Emerg. Res. Electron., Comput. Sci. Technol. (ICERECT)*, Mandya, India, Dec. 2015, pp. 132–136.
- [64] F. Jimenez and R. Torres, "Building an IoT-aware healthcare monitoring system," in *Proc. 34th Int. Conf. Chilean Comput. Sci. Soc. (SCCC)*, Santiago, Chile, Nov. 2015, pp. 1–4.
- [65] B. Negash et al., "Leveraging fog computing for healthcare IoT," in Fog Computing in the Internet of Things. Cham, Switzerland: Springer, 2018, pp. 145–169.
- [66] K. Rajeswari, N. Vivekanandan, P. Amitaraj, and A. Fulambarkar, "A study on redesigning modern healthcare using Internet of Things," in *Healthcare Systems Management: Methodologies and Applications*. Singapore: Springer, 2018, pp. 59–69.
- [67] Z. A. Khan and A. Samad, "A study of machine learning in wireless sensor network," *Int. J. Comput. Netw. Appl.*, vol. 4, no. 4, pp. 105–112, 2017.
- [68] J. Burrell, "How the machine 'thinks': Understanding opacity in machine learning algorithms," *Big Data Soc.*, vol. 3, pp. 1–12, Jun. 2016.
- [69] P. Domingos, "A few useful things to know about machine learning," *Commun. ACM*, vol. 55, no. 10, p. 78, Oct. 2012.
- [70] M. Monteiro, A. C. Fonseca, A. T. Freitas, T. P. Melo, A. P. Francisco, J. M. Ferro, and A. L. Oliveira, "Using machine learning to improve the prediction of functional outcome in ischemic stroke patients," *IEEE/ACM Trans. Comput. Biol. Bioinf.*, vol. 15, no. 6, pp. 1953–1959, Nov. 2018.
- [71] R. Chen and E. H. Herskovits, "Machine-learning techniques for building a diagnostic model for very mild dementia," *NeuroImage*, vol. 52, no. 1, pp. 234–244, Aug. 2010.
- [72] W. Raghupathi and V. Raghupathi, "Big data analytics in healthcare: Promise and potential," *Health Inf. Sci. Syst.*, vol. 2, no. 1, pp. 2–10, Dec. 2014.
- [73] M. Hauskrecht, I. Batal, M. Valko, S. Visweswaran, G. F. Cooper, and G. Clermont, "Outlier detection for patient monitoring and alerting," *J. Biomed. Informat.*, vol. 46, no. 1, pp. 47–55, Feb. 2013.
- [74] M. S. Kohn, J. Sun, S. Knoop, A. Shabo, B. Carmeli, and D. Sow, "IBM's health analytics and clinical decision support," *Yearbook Med. Informat.*, vol. 23, no. 1, pp. 154–162, Aug. 2014.
- [75] R. Bhardwaj, A. R. Nambiar, and D. Dutta, "A study of machine learning in healthcare," in *Proc. IEEE 41st Annu. Comput. Softw. Appl. Conf.* (COMPSAC), Turin, Italy, vol. 2, Jul. 2017, pp. 236–241.
- [76] J. Wu, J. Roy, and W. F. Stewart, "Prediction modeling using EHR data: Challenges, strategies, and a comparison of machine learning approaches," *Med. Care*, vol. 48, pp. S106–S113, Jun. 2010.
- [77] M. Motwani, "Machine learning for prediction of all-cause mortality in patients with suspected coronary artery disease: A 5-year multicentre prospective registry analysis," *Eur. Heart J.*, vol. 38, pp. 500–507, Feb. 2017.
- [78] J. Stausberg and M. Person, "A process model of diagnostic reasoning in medicine," *Int. J. Med. Informat.*, vol. 54, no. 1, pp. 9–23, Apr. 1999.
- [79] B. Zupan, J. A. Halter, and M. Bohanec, "Qualitative model approach to computer assisted reasoning in physiology," in *Proc. Intell. Data Anal. Med. Pharmacol.-(IDAMAP)*, 1998, pp. 1–7.
- [80] A. d'Avila Garcez, M. Gori, P. Hitzler, and L. C. Lamb, "Neural-symbolic learning and reasoning (dagstuhl seminar 14381)," *Dagstuhl Rep.*, vol. 4, pp. 50–84, Sep. 2015.
- [81] R. Gupta, M. Mitra, and J. Bera, ECG Acquisition and Automated Remote Processing. New Delhi, India: Springer, 2016.
- [82] D. T. Hau and E. W. Coiera, "Learning qualitative models of dynamic systems," *Mach. Learn.*, vol. 26, pp. 177–211, Feb. 1997.
- [83] V. Chaurasia and S. Pal, "Data mining techniques: To predict and resolve breast cancer survivability," *Int. J. Comput. Sci. Mobile Comput.*, vol. 3, pp. 10–22, Jan. 2014.
- [84] J. R. Sutton, R. Mahajan, O. Akbilgic, and R. Kamaleswaran, "PhysOnline: An open source machine learning pipeline for real-time analysis of streaming physiological waveform," *IEEE J. Biomed. Health Informat.*, vol. 23, no. 1, pp. 59–65, Jan. 2019.
- [85] E. Sejdic and T. H. Falk, Signal Processing and Machine Learning for Biomedical Big Data. Boca Raton, FL, USA: CRC Press, Jul. 2018, p. 624, doi: 10.1201/9781351061223.
- [86] K. López-de-Ipiña and C. Laske, "Editorial: Advanced methods of biomedical signal processing for early detection of Alzheimer's disease," *Current Alzheimer Res.*, vol. 14, no. 9, pp. 914–915, Aug. 2017.

- [87] P. Kukharchik, D. Martynov, I. Kheidorov, and O. Kotov, "Vocal fold pathology detection using modified wavelet-like features and support vector machines," in *Proc. 15th Eur. Signal Process. Conf.*, Poznań, Poland, Sep. 2007, pp. 2214–2218.
- [88] T. Dubuisson, T. Dutoit, B. Gosselin, and M. Remacle, "On the use of the correlation between acoustic descriptors for the Normal/Pathological voices discrimination," *EURASIP J. Adv. Signal Process.*, vol. 2009, no. 1, pp. 1–19, Dec. 2009.
- [89] C. Fredouille, G. Pouchoulin, J.-F. Bonastre, M. Azzarello, A. Giovanni, and A. Ghio, "Application of Automatic Speaker Recognition techniques to pathological voice assessment (dysphonia)," in *Proc. Eur. Conf. Speech Commun. Technol. (Eurospeech)*, Lisboa, France, 2005, pp. 149–152.
- [90] J. Wang and C. Jo, "Performance of Gaussian mixture models as a classifier for pathological voice," in *Proc. 11th Austral. Int. Conf. Speech Sci. Technol.*, vol. 107, 2006, pp. 122–131.
- [91] T. A. Mesallam, M. Farahat, K. H. Malki, M. Alsulaiman, Z. Ali, A. Al-Nasheri, and G. Muhammad, "Development of the arabic voice pathology database and its evaluation by using speech features and machine learning algorithms," *J. Healthcare Eng.*, vol. 2017, pp. 1–13, Oct. 2017.
- [92] R. Amami and A. Smiti, "An incremental method combining density clustering and support vector machines for voice pathology detection," *Comput. Electr. Eng.*, vol. 57, pp. 257–265, Jan. 2017.
- [93] P. Harar, J. B. Alonso-Hernandezy, J. Mekyska, Z. Galaz, R. Burget, and Z. Smekal, "Voice pathology detection using deep learning: A preliminary study," in *Proc. Int. Conf. Workshop Bioinspired Intell. (IWOBI)*, Funchal, Portugal, Jul. 2017, pp. 1–4.
- [94] G. Muhammad, M. Alhamid, M. Hossain, A. Almogren, and A. Vasilakos, "Enhanced living by assessing voice pathology using a co-occurrence matrix," *Sensors*, vol. 17, no. 2, p. 267, 2017.
- [95] T. Xie, X. Chen, J. Fang, H. Kang, W. Xue, H. Tong, P. Cao, S. Wang, Y. Yang, and W. Zhang, "Textural features of dynamic contrastenhanced MRI derived model-free and model-based parameter maps in glioma grading," *J. Magn. Reson. Imag.*, vol. 47, no. 4, pp. 1099–1111, Apr. 2018.
- [96] S. Minaee, A. Abdolrashidi, and Y. Wang, "Iris recognition using scattering transform and textural features," in *Proc. IEEE Signal Process. Signal Process. Edu. Workshop (SP/SPE)*, Salt Lake, UT, USA, Aug. 2015, pp. 37–42.
- [97] A. E. Fetit, J. Novak, A. C. Peet, and T. N. Arvanitis, "Three-dimensional textural features of conventional MRI improve diagnostic classification of childhood brain tumours," *NMR Biomed.*, vol. 28, no. 9, pp. 1174–1184, Sep. 2015.
- [98] D. Martínez, E. Lleida, A. Ortega, and A. Miguel, "Score level versus audio level fusion for voice pathology detection on the Saarbrücken voice database," in *Advances in Speech and Language Technologies for Iberian Languages*. Berlin, Germany: Springer, 2012, pp. 110–120.
- [99] G. Muhammad, M. Alsulaiman, Z. Ali, T. A. Mesallam, M. Farahat, K. H. Malki, A. Al-nasheri, and M. A. Bencherif, "Voice pathology detection using interlaced derivative pattern on glottal source excitation," *Biomed. Signal Process. Control*, vol. 31, pp. 156–164, Jan. 2017.
- [100] A. Al-nasheri, G. Muhammad, M. Alsulaiman, Z. Ali, T. A. Mesallam, M. Farahat, K. H. Malki, and M. A. Bencherif, "An investigation of multidimensional voice program parameters in three different databases for voice pathology detection and classification," *J. Voice*, vol. 31, no. 1, pp. 113.e9–113.e18, Jan. 2017.
- [101] F. Amara, M. Fezari, and H. Bourouba, "An improved GMM-SVM system based on distance metric for voice pathology detection," *Appl. Math. Inf. Sci.*, vol. 10, no. 3, pp. 1061–1070, May 2016.
- [102] H. Cordeiro, J. Fonseca, I. Guimaraes, and C. Meneses, "Voice pathologies identification speech signals, features and classifiers evaluation," in *Proc. Signal Process., Algorithms, Archit., Arrangements, Appl. (SPA)*, Poznań, Poland, Sep. 2015, pp. 81–86.
- [103] N. Souissi and A. Cherif, "Dimensionality reduction for voice disorders identification system based on mel frequency cepstral coefficients and support vector machine," in *Proc. 7th Int. Conf. Model., Identificat. Control (ICMIC)*, Sousse, Tunisia, Dec. 2015, pp. 1–6.
- [104] I. M. M. El Emary, M. Fezari, and F. Amara, "Towards developing a voice pathologies detection system," *J. Commun. Technol. Electron.*, vol. 59, no. 11, pp. 1280–1288, Nov. 2014.
- [105] V. Sellam and J. Jagadeesan, "Classification of normal and pathological voice using SVM and RBFNN," J. Signal Inf. Process., vol. 5, no. 1, pp. 1–7, 2014.

- [106] G. Daza-Santacoloma, J. D. Arias-Londoño, J. I. Godino-Llorente, N. Sáenz-Lechón, V. Osma-Ruíz, and G. Castellanos-Domínguez, "Dynamic feature extraction: An application to voice pathology detection," *Intell. Automat. Soft Comput.*, vol. 15, pp. 667–682, Mar. 2013.
- [107] J. I. Godino-Llorente, P. Gómez-Vilda, N. Sáenz-Lechón, M. Blanco-Velasco, F. Cruz-Roldán, and M. A. Ferrer, "Discriminative methods for the detection of voice disorders," in *Proc. ISCA Tutorial Res. Workshop (ITRW) Non-Linear Speech Process.*, Barcelona, Spain, 2005, pp. 158–167.
- [108] A. A. Dibazar and S. Narayanan, "A system for automatic detection of pathological speech," in *Proc. Conf. Signals, Syst., Comput.*, Asilomar, CA, USA, Oct. 2002, pp. 1–4.
- [109] Z. Ali, I. Elamvazuthi, M. Alsulaiman, and G. Muhammad, "Detection of voice pathology using fractal dimension in a multiresolution analysis of normal and disordered speech signals," *J. Med. Syst.*, vol. 40, no. 1, p. 20, Jan. 2016.
- [110] J. I. Godino-Llorente, P. Gomez-Vilda, and M. Blanco-Velasco, "Dimensionality reduction of a pathological voice quality assessment system based on Gaussian mixture models and short-term cepstral parameters," *IEEE Trans. Biomed. Eng.*, vol. 53, no. 10, pp. 1943–1953, Oct. 2006.
- [111] R. R. A. Barreira and L. L. Ling, "Kullback–Leibler divergence and sample skewness for pathological voice quality assessment," *Biomed. Signal Process. Control*, vol. 57, Mar. 2020, Art. no. 101697.
- [112] I. Hammami, L. Salhi, and S. Labidi, "Voice pathologies classification and detection using EMD-DWT analysis based on higher order statistic features," *IRBM*, to be published.
- [113] E. S. Fonseca, R. C. Guido, S. B. Junior, H. Dezani, R. R. Gati, and D. C. M. Pereira, "Acoustic investigation of speech pathologies based on the discriminative paraconsistent machine (DPM)," *Biomed. Signal Process. Control*, vol. 55, Jan. 2020, Art. no. 101615.
- [114] M. K. Reddy, P. Alku, and K. S. Rao, "Detection of specific language impairment in children using glottal source features," *IEEE Access*, vol. 8, pp. 15273–15279, 2020.
- [115] G. Solana-Lavalle, J.-C. Galán-Hernández, and R. Rosas-Romero, "Automatic parkinson disease detection at early stages as a pre-diagnosis tool by using classifiers and a small set of vocal features," *Biocybern. Biomed. Eng.*, vol. 40, no. 1, pp. 505–516, Jan. 2020.
- [116] M. Alhussein and G. Muhammad, "Automatic voice pathology monitoring using parallel deep models for smart healthcare," *IEEE Access*, vol. 7, pp. 46474–46479, 2019.
- [117] A. Rueda and S. Krishnan, "Augmenting dysphonia voice using Fourierbased synchrosqueezing transform for a CNN classifier," in *Proc. ICASSP IEEE Int. Conf. Acoust., Speech Signal Process. (ICASSP)*, Brighton, U.K., May 2019, pp. 6415–6419.
- [118] F. N. C. Kassim, V. Vijean, H. Muthusamy, R. Abdullah, and Z. Abdullah, "Mobile devices and apps for health care professionals: Uses and benefits," in *Proc. Int. Conf. Biomed. Eng.*, Penang Island, Malaysia, vol. 1372, Aug. 2019, pp. 1–6.
- [119] V. Mittal and R. K. Sharma, "Glottal signal analysis for voice pathology," in *Proc. 2nd Int. Conf. Innov. Electron., Signal Process. Commun. (IESC)*, Shillong, India, Mar. 2019, pp. 54–59.
- [120] S. U. Amin, M. S. Hossain, G. Muhammad, M. Alhussein, and M. A. Rahman, "Cognitive smart healthcare for pathology detection and monitoring," *IEEE Access*, vol. 7, pp. 10745–10753, 2019.
- [121] S. R. Kadiri and P. Alku, "Analysis and detection of pathological voice using glottal source features," *IEEE J. Sel. Topics Signal Process.*, early access, Dec. 6, 2019, doi: 10.1109/JSTSP.2019.2957988.
- [122] M. Alhussein and G. Muhammad, "Voice pathology detection using deep learning on mobile healthcare framework," *IEEE Access*, vol. 6, pp. 41034–41041, 2018.
- [123] A. Al-Nasheri, G. Muhammad, M. Alsulaiman, Z. Ali, K. H. Malki, T. A. Mesallam, and M. Farahat Ibrahim, "Voice pathology detection and classification using auto-correlation and entropy features in different frequency regions," *IEEE Access*, vol. 6, pp. 6961–6974, 2018.
- [124] L. Verde, G. De Pietro, and G. Sannino, "Voice disorder identification by using machine learning techniques," *IEEE Access*, vol. 6, pp. 16246–16255, 2018.
- [125] K. Ezzine and M. Frikha, "Investigation of glottal flow parameters for voice pathology detection on SVD and MEEI databases," in *Proc. 4th Int. Conf. Adv. Technol. Signal Image Process. (ATSIP)*, Sousse, Tunisia, Mar. 2018, pp. 1–6.
- [126] I. Hammami, L. Salhi, and S. Labidi, "Pathological voices detection using support vector machine," in *Proc. 2nd Int. Conf. Adv. Technol. Signal Image Process. (ATSIP)*, Monastir, Tunisia, Mar. 2016, pp. 662–666.

- [127] M. S. Hossain and G. Muhammad, "Healthcare big data voice pathology assessment framework," *IEEE Access*, vol. 4, pp. 7806–7815, 2016.
- [128] N. Souissi and A. Cherif, "Speech recognition system based on shortterm cepstral parameters, feature reduction method and artificial neural networks," in *Proc. 2nd Int. Conf. Adv. Technol. Signal Image Process.* (ATSIP), Monastir, Tunisia, Mar. 2016, pp. 667–671.
- [129] C. R. Francis, V. V. Nair, and S. Radhika, "A scale invariant technique for detection of voice disorders using modified Mellin transform," in *Proc. Int. Conf. Emerg. Technol. Trends (ICETT)*, Kollam, India, Oct. 2016, pp. 1–6.



FAHAD TAHA AL-DHIEF (Student Member, IEEE) was born in Amarah, Iraq. He received the B.S. degree in software engineering from Imam Ja'afar Al-Sadiq University, Iraq, in 2013, and the M.S. degree in computer science from Universiti Kebangsaan Malaysia, Malaysia, in 2016. He is currently pursuing the Ph.D. degree with the School of Electrical Engineering, Universiti Teknologi Malaysia, Malaysia. His research interests are sensor networks, routing protocols, mobile

ad-hoc networks, social networks, the Internet of Things, machine learning, artificial neural networks, deep learning, and location-based service. He is a member of the IEEE Communications Society.



NURUL MU'AZZAH ABDUL LATIFF (Senior Member, IEEE) received the Bachelor of Engineering degree in electrical-telecommunication from UTM, in 2002, and the M.Sc. degree in communication and signal processing and the Ph.D. degree in wireless communication engineering from Newcastle University, in 2003 and 2008, respectively. She is currently an Associate Professor with the Division of Communication Engineering, School of Electrical Engineering, Universiti

Teknologi Malaysia (UTM). Her research interests include wireless sensor networks, mobile ad hoc networks, cognitive radio, the Internet of Things, and artificial intelligence. She is a member of the IEEE Communications Society and the Vehicular Technology Society, besides being a certified Chartered Engineer from IET.



NIK NOORDINI NIK ABD. MALIK (Member, IEEE) received the bachelor's degree in electrical engineering (telecommunication) from Universiti Teknologi Malaysia (UTM), in 2003, the Master of Engineering (M.Eng.) degree in radio frequency (RF) and microwave communication engineering from The University of Queensland, Australia (UQ), in 2005, and the Doctor of Philosophy (Ph.D.) degree in electrical engineering from Universiti Teknologi Malaysia (UTM), Malaysia,

in 2013. After that, she worked as a Radio Frequency R&D Electrical Engineer at Motorola Technology Ptd. Ltd., Penang, Malaysia. She is currently a Senior Lecturer with the School of Electrical Engineering, UTM. Her research interests include wireless sensor networks, distributed beamforming, and meta-heuristic algorithms.



NASEER SABRI SALIM received the B.Sc. and M.Sc. degrees in computer engineering within intelligent computer vision field. His Ph.D. degree concerns the development of a novel intelligent WSN. He is currently a Faculty Member with the Computing and Information Technology, Sohar University, Sohar, Oman. His main research area targeting on building a new intelligent wireless sensor network based on an artificial intelligence paradigm. He has also authored or coauthored

more than 45 technical articles of internationally cited journals. His research interests are on automation, smart wireless sensor network solutions, embedded system design, and computer vision. He is also serving as reviewers for various international publications including top-ten ranked journals.



MUSATAFA ABBAS ABBOOD ALBADR received the bachelor's degree from the Department of Computer Science, College of Education, University of Basrah, Iraq, in 2011, and the M.S. degree in computer science from University Kebangsaan Malaysia, Malaysia, in 2017. He is currently pursuing the Ph.D. degree with the Faculty of Information Science and Technology, University Kebangsaan Malaysia, Malaysia. His research interests are machine learning, neural

networks, speech processing, data mining, parallel processing, and metaheuristic algorithms.



MARINA MAT BAKI graduated from the Faculty of Medicine, Universiti Kebangsaan Malaysia, Malaysia. In 1997, she started her medical career with the Ministry of Health of Malaysia and the Otorhinolaryngology Training at the Universiti Kebangsaan Malaysia Medical Centre (UKMMC), in 2002. She completed the ORL Training four years later, in 2006. She is currently the Professor of Laryngology of Department of UKMMC, Malaysia. Her special interest is laryngotracheal

stenosis, laryngeal paralysis, voice disorders, phonosurgery, and sleep apnoea. Since then, she has been inspiring and trained by Prof. Dato' Dr. Abdullah Sani Mohamed to take up laryngology. Subsequently she was awarded Ph.D. in laryngology of the University College London, in November 2014. During the Ph.D. Program, she received the certificate of subspeciality training in laryngology when she worked with distinguished laryngologists and voice surgeons: Prof. M. Birchall, Dr. J. Rubin, and Dr. G. Sandhu at the Royal National Throat Nose and Ear Hospital and Charing Cross Hospital in London, U.K., from July 2010 to May 2014.



MAZIN ABED MOHAMMED received the B.Sc. degree in computer science from the College of Computer, University of Anbar, Anbar, Iraq, in 2008, the master's degree in IT from Universiti Tenaga Nasional Malaysia, in 2011, and the Ph.D. degree from Universiti Teknikal Malaysia Melaka, Malaysia, in 2018. He is currently working as a Teacher at the Department of Planning and Follow Up, University of Anbar. He teaches a variety of university courses in computer science, such as

operating systems, database design, mobile systems programing, software project management, web technologies, and software requirements and design. His areas of research interest include artificial intelligence, medical image processing, machine learning, computer vision, computational intelligence, the Internet of Things (IoT), biomedical computing, bio-informatics, and fog computing.